**Author's response to reviews**

**Title:** The impact of age on post-operative outcomes of colorectal cancer patients undergoing surgical treatment

**Authors:**

Tadas Latkauskas (Tadas.Latkauskas@kmuk.lt)  
Giedre Rudinskaite (grudinsk@yahoo.com)  
Juozas Kurtinaitis (jkurtinaitis@mail.lt)  
Rasa Janciauskiene (Rasa.Janciauskiene@kmuk.lt)  
Algimantas Tamelis (kp@kmu.lt)  
Zilvinas Saladzinskas (kp@kmu.lt)  
Dainius Pavalkis (dainius@kmu.lt)

**Version:** 5  **Date:** 27 November 2005

**Author's response to reviews:**

1. Title and Abstract are identical in the manuscript and submission system.

2. Authors list and affiliations - changed.

3. Keywords section - removed.


5. Headings - sorry, but I did not understand this point.


7. Figures - removed, titles of the figures are in the bottom of the manuscript.